Author's response to reviews

Title: Risk factor analysis and spatiotemporal CART model of cryptosporidiosis in Queensland, Australia

Authors:

Wenbiao Hu (w.hu@sph.uq.edu.au)
Kerrie Mengersen (k.mengersen@qut.edu.au)
Shilu Tong (s.tong@qut.edu.au)

Version: 5 Date: 18 October 2010

Author's response to reviews: see over
Dear Editor,

Thank you very much for your letter of the 13\textsuperscript{th} October 2010 on our paper. Please find enclosed a copy of our revised manuscript with the following changes, according to your suggestions:

1. We have changed the numbers of figures as you suggested. Any references to the figures in the manuscript were changed.

2. We have minimised white space around the figures as you suggestion.

3. We have carefully check typographical errors for the manuscript as your suggestions.

Thank you very much for giving us the opportunity to revise this paper.

Yours sincerely,

Wenbiao Hu*                         Kerrie Mengersen                     Shilu Tong

*Corresponding author: Dr. Wenbiao Hu. Telephone: +61-7-3138 8295; Facsimile: +61-7-3138 3130; Email: w2.hu@qut.edu.au